Author’s response to reviews

Title: Inequities in incidence, morbidity and expenditures on prevention and treatment of malaria in southeast Nigeria

Authors:

Nkolika P Uguru (nk_uguru@yahoo.com)
Obinna E Onwujekwe (onwujekwe@yahoo.co.uk)
Benjamin S Uzochukwu (bsuczochukwu@yahoo.com)
Godwin C Igiliegbe (igiliegbechuks@yahoo.com)
Soludo B Eze (soludo2001@yahoo.co.uk)

Version: 7 Date: 29 August 2009

Author’s response to reviews:

DEPARTMENT OF PHARMACOLOGY AND THERAPEUTICS
COLLEGE OF MEDICINE
UNIVERSITY OF NIGERIA
PMB 01129, ENUGU, NIGERIA
29/08/09

Dear Editor,

Resubmission of a manuscript for possible publication

We hereby resubmit a manuscript entitled Inequities in incidence, morbidity and expenditures on prevention and treatment of malaria in southeast Nigeria

The neccessary formatting and grammar corrections have been made to this paper, I hope it is now acceptable for publication in your journal.

This paper is not being considered elsewhere for publication.

Yours sincerely,

Nkoli Uguru